Author’s response to reviews

Title: Patterns and trends of medical student research

Authors:

Dakshitha P Wickramasinghe (dakshithaw@hotmail.com)
Chamila S Perera (chamilasudarshi@gmail.com)
Supun Senaratne (supunmfc05@gmail.com)
Dharmabandhu N Samarasekera (samarasekera58@yahoo.co.uk)

Version: 4 Date: 14 December 2013

Author’s response to reviews:

Reviewers have accepted the changes done to the article previously. In addition, as requested by the Editorial Board, we have made the following changes.

1. We have checked the article to see whether it conforms to the formatting check list.
2. A section on competing interest was added (highlighted in yellow)
3. A section on acknowledgements was added (highlighted in yellow)